Author's response to reviews

Title: Neuroendocrine tumour (small cell/oat cell carcinoma) of the breast, a treatment between breast and lung: a case report

Authors:

Stefania Nicoletti (snicoletti23@libero.it)
Maximilian Papi (papi1@interfree.it)
Debora Canuti (dcanuti@auslrn.net)
Fabrizio Drudi (f.drudi1@virgilio.it)
Emiliano Tamburini (bagabbrk@libero.it)
Cinzia Possenti (cinzia.possenti@libero.it)
Enzo Pasquini (enzo.pasquini@auslrn.net)
Manuela Fantini (fantamanu77@libero.it)
Massimo Brisigotti (brisigotti@auslrn.net)
Alberto Ravaioli (aravaiol@auslrn.net)

Version: 8 Date: 19 February 2010

Author's response to reviews:

Dear Editorial Board,

We uploaded a revised version of the manuscript according to your requirements as listed below:

1) We wrote exactly the list of the authors in correct style and order.
2) We added the authors' contributions.
3) We wrote in full each affiliation.
4) We stated the consent as requested.
5) We corrected the abstract DISCUSSION AND CONCLUSIONS in CONCLUSIONS.
6) We put the consent section before competing interest section.
7) We completed the authors' references.
8) We uploaded the figure separately.
9) We checked the typography of the manuscript.
10) We returned the revised version before 26th Feb. 2010.